Author’s response to reviews

Title: Healthy Living after Cancer: A dissemination and implementation study evaluating a telephone-delivered healthy lifestyle program for cancer survivors

Authors:

Elizabeth Eakin (e.eakin@sph.uq.edu.au)
Sandra Hayes (sc.hayes@qut.edu.au)
Marion Haas (Marion.Haas@chere.uts.edu.au)
Marina Reeves (m.reeves@sph.uq.edu.au)
Janette Vardy (janette.vardy@sydney.edu.au)
Frances Boyle (franb@bigpond.net.au)
Janet Hiller (jhiller@swin.edu.au)
Gita Mishra (g.mishra@sph.uq.edu.au)
Ana Goode (a.goode@sph.uq.edu.au)
Michael Jefford (Michael.Jefford@petermac.org)
Bogda Koczwara (Bogda.Koczwara@health.sa.gov.au)
Christobel Saunders (christobel.saunders@uwa.edu.au)
Wendy Demark-Wahnefried (demark@uab.edu)
Kerry Courneya (kerry.courneya@ualberta.ca)
Kathryn Schmitz (schmitz@mail.med.upenn.edu)
Afaf Girgis (afaf.girgis@unsw.edu.au)
Kate White (kate.white@sydney.edu.au)
Kathy Chapman (kathyc@nswcc.org.au)
Anna Bolton (Anna.Bolton@cancervic.org.au)
Katherine Lane (Katherine.Lane@cancervic.org.au)
Author’s response to reviews:

Dear Dr Solera,

Thank you for your email regarding our manuscript, "Healthy Living after Cancer: A dissemination and implementation study evaluating a telephone-delivered healthy lifestyle program for cancer survivors" (BCAN-D-15-00468).

Please find our responses to your queries below:

1. Copies of all ethical approval and funding approval documents have been forwarded to BMCSeriesEditorial@biomedcentral.com as requested. This has also been noted in the amended cover letter for our submission.

2. Healthy Living after Cancer received funding from the Australian National Health and Medical Research Council (NHMRC), and as such has been peer reviewed. This has been noted in the amended cover letter. In addition, details of this funding have now been added to the "Competing Interests" section of the manuscript.

3. Healthy Living after Cancer is ongoing - data collection began in June 2015 and will continue until December 2017. This has been noted in the amended cover letter.

4. As the study is ongoing, no publications containing the results of the study have already been published or submitted to any journal. This has been noted in the amended cover letter.

Kind regards.